Author’s response to reviews

Title: Case report: New development of fibrosing interstitial lung disease triggered by HIV-related pneumocystis pneumonia.

Authors:
Tetsuya Suzuki (tesuzuki@hosp.ncgm.go.jp)
Yukiko Shimoda (yushimoda@hosp.ncgm.go.jp)
Katsuji Teruya (kteruya@acc.ncgm.go.jp)
Hiroyuki Gatanaga (higatana@acc.ncgm.go.jp)
Yoshimi Kikuchi (yoshik@acc.ncgm.go.jp)
Shinichi Oka (oka@acc.ncgm.go.jp)
Koji Watanabe (kwatanab@acc.ncgm.go.jp)

Version: 1 Date: 01 Mar 2018

Author’s response to reviews:

Dear editorial office,

We completed CARE checklist, and made editorial changes of our manuscript as you indicated. Please see attached files, and kindly let us know if we need to modify more.

Corresponding author

Koji Watanabe, AIDS Clinical Center, National Center for Global Health and Medicine, Tokyo, Japan.